Author’s response to reviews

Title: Foodborne norovirus outbreak: the role of an asymptomatic food handler

Authors:

Irene Barrabeig (ibarrabeig@catsalut.cat)
Ariadna Rovira (arovira@catsalut.cat)
Javier Buesa (javier.buesa@uv.es)
Rosa Bartolomé (rbartolome@vhebron.net)
Rosa Pintó (rpinto@ub.edu)
Àngela Domínguez (angela.dominguez@ub.edu)

Version: 6 Date: 1 June 2010

Author’s response to reviews: see over
Ms Roxane Rajabi
The BioMed Central Editorial Team

Dear Ms. Roxane Rajabi,

Please find enclosed in Methods the paragraph explaining why it was not necessary in this study to ask for an informed consent: *The investigation was exempted from ethical committee approval as it was part of the national public health response to the outbreak.*

You will see in red the sentence made in the last manuscript I am sending you.

Sincerely yours,

Angela Dominguez
Department of Public Health.
University of Barcelona

Barcelona, 1st June 2010